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Device Service Record Form for TB9ACC012
	Date
	Comments

(Include the following in each DSR entry: What, Where, Why)
	Info Provided By

	2/9/09
	This is the DSR for the 9-cell cavity TB9ACC012.
Cavity purchased from ACCEL on PO 571761, and routed into TD under Routing Form 87897.

QC Inspection Summary (all travelers in Vector):

i) Visual Inspection (TR-333876): completed 3/2008.

ii) CMM Measurements (TR-333897): completed without 180-degree orientation measurements 4/2008.

iii) RF Measurements (TR-333873): completed 3/2008.
iv) Leak check (TR-333878): not completed.
	Jamie Blowers, Nandhini Dhanaraj

	4/7/2009
	JLAB is going to dis-assemble and HPR with burst disk.
	Genfa Wu (CMG)

	4/22/2009
	Email from Jim Follkie to CMG et al.:
A12 shipped yesterday. ETA to Fermi is 3 to 5 days.

Attached is the bill of lading.
	Camille Ginsburg

	5/7/2009
	Received at Fermilab 4/27/09.
	John Zweibohmer

	5/15/2009
	RF test 5/15 at Fermilab: max gradient 39.1MV/m, with Q0 at max gradient 7.18 E9; limitation: cryogenic system pumping capacity. No field emission or radiation observed at any point during the run. 

Ozelis test note: document
	Camille Ginsburg

	5/27/2009
	Has been moved to MP9 for processing and use in CM2.
	John Zweibohmer

	6/2/2009
	Email from Rongli to CMG 6/2:
Final RF test max gradient (JLab): 35.1 MV/m (1st pass); 39.9 MV/m (after re-EP)

Replaces 7/7/2008 test at 36.6 MV/m and 7/21/2008 test at 37.0 MV/m.
	Camille Ginsburg

	6/2/2009
	Moved to IB#4 for CMM measurement.
	John Zweibohmer

	6/5/2009
	Cavity moved back to MP9 for installation of RF flanges to allow for Timer to measure the cavity during welding operations.  No CMM measurements were performed at this time, but will be after ebeam welding on 6/18 & 6/19.
	John Zweibohmer

	6/23/2009
	Jim Follkie EP parameter spreadsheet, including TB9ACC012 is here.
	Camille Ginsburg

	7/14/2009
	Mark Champion: This cavity got a hole blasted in its end cell in an electron beam welding accident at Sciaky while the titanium transition rings were being welded on, in preparation for helium vessel welding. Pictures
	Camille Ginsburg

	10/15/09
	Per Timergali – Cavity damaged during electron beam welding not a qualified cavity, still under review 10/13/09.
	Jim Rife

	12/15/09
	Location verification.  Cavity at ICB 12/15/09.
	Jim Rife

	12/30/09
	Cavity being sent to AES on MMR 146379. Transportation will pick-up from ICB. 
	L. Zimmerman

	11/15/2010
	Cavity location verified ICB after return from repair at AES.
	Camille Ginsburg

	1/6/2011
	Cavity ready to be picked up after optical inspection.
	E. Toropov

	1/10/2011
	Cavity moved from ICB to IB4 for inspection
	L. Zimmerman / R. Espinoza

	Jan/13/2011
	Cavity leak checked by Derek Plant
	R. Riley

	Jan/19/2011
	Cavity measured on CMM.  Cavity returned to crate on IB4 floor.
	R. Riley

	Jan/24/2011
	Cavity moved to IB4 RF Lab for RF QC
	T. Khabiboulline

	Jan/25/2011
	Cavity moved out after RF QC and returned to crate on IB4 floor.
	T. Khabiboulline

	1/27/11
	Cavity moved to ICPA Tumbling area to be prepped and loaded.
	G.Steuer

	1/28/11
	Began Tumbling process on TB9ACC012
	G.Steuer

	2/3/11
	Cavity moved to ICB for Optical Inspection 
	G.Steuer

	2/3/11
	Cavity moved to ICPA Tumbling area for more tumbling
	G.Steuer

	1/27/11 
	Cavity loaded with ACT  Type M tumbling media 12.5L (17895g) with 
Ultra pure water (5021g) and TS compound (75ml) cavity placed into tumbler @1715 did not start tumbler.
	dburk

	4/4/11
	Cavity completed tumbling process. Cavity tumbled with 5 different medias. Tumbled with offset pyramids, ½ in cones, 400 mesh alumina, 800 mesh alumina, and colloidal silica.  Surface very shiny but still some scratches on surface.
	C. Cooper

	4/4/11
	Cavity moved to optical inspection in ICB.
	C. Cooper

	4/7/11
	The inspection is finished cavity is ready to be picked up
	E. Toropov

	4/7/11
	Cavity moved to IB4 per Timer
	JB/RE

	4/7/2011
	Cavity moved to IB4 RF Lab for measurements.
	T. Khabiboulline

	4/8/2011
	RF measurements done. F=1296.15 MHz, field flatness 64%. Cavity moved back to crate in IB4 main floor.
	T. Khabiboulline

	4/28/11
	Cavity moved from IB4 to ICB for storage
	JB/RE

	5/11/11
	Cavity at ICB.  Next; 30-40um EP, 800C bake, tuning, 10um EP, vertical test prep, vertical test – on hold.  See Ginsburg Cavity Scheduling Mtg. 4-May-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 4-May-2011.

	5/16/11
	Cavity moved to IB4-ICPA tumbling area to be prepped for 5th step.
	George Steuer

	5/16/11
	Cavity was HPR in IB4 
	dburk

	5/25/11
	Cavity loaded with ray-cubes and colloidal silica .04 micron (soaked overnight) started tumbling to be tumbled as long as possible
	Burk

	5/29/11
	Tumbling over the long weekend DB checked in the morning and found that the tumbler was stopped I tried to reset tumbler this did not work. Called CC and we removed both cavities (tumbling AES006 at the same time) removed media and rinsed
	Burk

	5/31/11
	Cavity was HPR at IB4
	Burk

	6/1/11
	Cavity placed in crate in IB4
	Burk

	6/1/2011
	Cavity transported from IB4 to ANL
	LZ

	    6/3/2011
	Cavity transported from ANL to IB4 per Damon Bice @ 1130 hours
	Rick Espinoza

	6/6/11
	Next: 20um EP, 800C bake (3hr), tuning, 10 um EP, vertical test prep, vertical test.  See Ginsburg Cavity Scheduling Mtg. 1-June-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 1-June-2011.

	6/8/11
	Baked at 800c w/ 3 hour hold and placed on cart for Tuning near tuning area
	Damon Bice

	06/08/2011
	Moved for RF tuning. FF=65.4%, F=1296.15 MHz
	T. Khabiboulline

	06/09/2011
	Cavity tuned, FF=99.0%, F=1297.34 MHz. Moved to transport container in IB4 main floor.
	T. Khabiboulline

	6/9/2011
	Cavity transported from IB4 to ANL per Damon Bice @ 1500 hours
	Rick Espinoza

	6/9/11
	Cavity received 10um EP and is being prepared for VTS
	Damon Bice

	6/20/11
	The Lower (FPC End) HOM can flange has now leaked twice consecutively. The lower 1.5 cells are the cells that were replaced. Further investigation needed into cause of leak.
	Damon Bice

	6/30/2011
	Cavity transported from ANL to IB1 per Damon Bice @ 1300 hours
	Rick Espinoza

	7/1/11
	Cavity received at IB1
	D. Marks

	7/9/11
	Cavity inserted into VTS Dewar after 48hr/120C bakeout
	J. Ozelis

	7/11/11
	Completed test of TB9ACC012. Cavity was quench-limited to 34.5 MV/m, with a Q at quench of 1.0e10. Low field Q was 2.4E10. Cavity was FE-free. Mode mixing observed form ~28-30 MV/m onward. Mode 

measurements indicate cells 1, 2, 8, and  9 are limited to 34.5 - 35.0 MV/m, cells 4 & 5 reach slightly higher gradients (36.5 MV/m) and cells 3, 5, and 7 reach in excess of 40 MV/m (RF power limited). Performed Q measurements as a fn of T; residual resistance appears to be about 5 nano-Ohms (typical good 9-cell cavity value). 
	J. Ozelis

	7/18/11
	Cavity is under vacuum, crated, and ready for pick-up
	D. Marks

	7/13/11
	Cavity at IB1.  VT 11-July-2011.  34.5 MV/m Q0=1E10 (quench), no measured FE.  See Ginsburg Cavity Scheduling Mtg. 13-July-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 13-July-2011

	7/20/11
	Cavity moved from IB1 to ICB racks
	LZ / JB

	8/30/11
	Cavity at ICB.  Next: dress it.  See Ginsburg Cavity Scheduling Mtg. 24-Aug-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 24-Aug-2011

	12/19/11
	Next: dressing- Dress after TB9AES012
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 14-Dec-2011

	8/22/12
	Cavity location verification.  Cavity @ ICB.
	Jim Rife

	4/10/13
	Cavity location verification.  Cavity at ICB.
	Jim Rife

	12/17/13
	VTS test for studying magnetic field in the dewar with different cooldown patterns and its effect on Rres
	Alex Melnychuk

	12/19/13
	VTS test (cavity remained in the dewar since previous test) 

Copy of e-log entry by Dmitri:

After "slow" cooldown the cavity still had a high initial Q0 but in the range 10 MV/m  to 15 MV/m it went down by about factor of 2 (see attached).   

Cavity quenched at ~32/34 MV/m. The quench was facilitated by excitation of a parasitic mode (this is common for ACCEL / RI cavities). Note there we no X-ray observed at all.  

We would like to warm up the cavity to above Tc and cool it back down with a normal procedure to see if the Q switch-over behavior persist or the cavity returns to its usual behavior.
	Alex Melnychuk

	1-10-14
	Tested @ VTS and transported to ICB
	Damon Bice

	1/14/14
	Cavity location verification.  Cavity at ICB.
	Jim Rife

	2-17-2014
	Cavity moved from ICB to IB4 for High Q0 bake—N2 only
	Rick Espinoza

	2-21-2014
	VTS prep at IB4 cleanroom—no EP after N2 heat treatment
	Allan Rowe

	3-3-2014
	Cavity moved from IB4 to IB1 for VT
	Rick Espinoza

	3-6-2014
	Cavity moved from IB1 to ICB
	Rick Espinoza

	3-7-2014
	Cavity transported from ICB to ANL
	Jim Rife

	3/7/14
	Cavity received at ANL
	Chris Baker

	3/11/2014
	Cavity received 7µm EP at ANL
	Ryan Murphy

	3/14/14
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	3/17/14
	Cavity transported to IB1
	Chris Baker

	3-20-2014
	Cavity moved from IB1 to ICB
	Rick Espinoza

	3-21-2014
	Cavity moved from ICB to IB4 RF Lab
	Rick Espinoza

	3-24-2014
	Cavity QC before tuning: Freq=1297.1 MHz Q=9121 FF=97%

After tuning: Freq=1297.35 MHz Q=9160 FF=98%. Cavity located in IB4 RF Lab.
	P. Berrutti

	3-31-2014
	Cavity moved from IB4 RF lab to Optical Inspection Lab
	Rick Espinoza

	4-1-2014
	Cavity was optically inspected
	Rick Espinoza

	4/2/14
	Cavity transported from ICB to ANL.
	Jim Rife

	04/02/14
	Cavity received at ANL
	Chris Baker

	4/3/2014
	Cavity received 5µm EP at ANL
	Ryan Murphy

	4/8/14
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	4/9/14
	Cavity transported from ANL to ICB.
	Jim Rife

	4/10/14
	Cavity transported from ICB to IB1 for testing.
	Jim Rife

	4/11/14
	VTS test results from T&I e-log.   55642
	Jim Rife

	4/14/14
	Cavity transported from IB1 to ICB.
	Jim Rife

	4/17/14
	Cavity transported from ICB to ANL.
	Jim Rife

	4/17/14
	Cavity received at ANL
	Chris Baker

	4/18/14
	Cavity transported from ANL to Fermi site 38 for shipment to JLab.
	Jim Rife

	4/21/14
	Cavity arrived at JLab 4/21/14.
	Jim Rife

	5/5/14
	Cavity transported from site 38 to IB1 for testing.
	Jim Rife

	5-13-14
	Cavity moved from IB1 to IB4
	Rick Espinoza

	5/14/14
	RF tuning completed: Freq=1297.33 MHz, FF=99%, Ecc=0.5 mm. Cavity located at IB4.
	PB

	5/16/14
	Cavity transported from IB4 to MP9.
	Jim Rife

	9/30/14
	Cavity delivered from Cornell to ICB.
	Jim Rife

	10/15/17
	Cavity moved from ICB to IB4 
	Bgolden 

	10/20/14
	Cavity assembled and under vacuum; moved to IB4 furnace area 
	Bgolden 

	10-21-2014
	Cavity moved from IB4 to IB1 and mounted to top plate #3
	Rick Espinoza

	10/31/14
	Test results from T&I e-log.  58292
	Jim Rife

	11/7/2014 
	Cavity dropped off by (J. Rife) from IB1 testing.
	Mark Chlebek

	11/18/14
	Cavity back from HTS
	Mark Chlebek

	2/13/15
	Cavity location verification.  Cavity at MP9.
	Jim Rife

	3/6/15
	Cavity transported from MP9 to IB1 for testing.
	Jim Rife

	3/16/15
	Cavity transported IB1 to IB4; cavity was re-evacuated 
	Bgolden 

	3/17/15
	Cavity transported from IB4 to IB1 for testing 
	Bgolden 

	3/18/15
	Test results from T&I e-log.  60172
	Jim Rife

	3/27/15
	Test results from T&I e-log.   60257
	Jim Rife

	3/27/15
	Test results from T&I e-log.   60258
	Jim Rife

	4-6-2015
	Cavity moved from IB1 to ICB for storage
	Rick Espinoza

	4/10/15
	Cavity transported from ICB to IB4 for disassembly.
	Jim Rife

	6/22/15
	Location update to ICB
	Damon Bice

	11/04/15
	Cavity transported to ANL
	Chris Baker

	11/04/15
	Cavity received at ANL
	Chris Baker

	11/06/15
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	11/9/15
	Cavity transported from ANL to MP9 for HTS.
	Jim Rife
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